

GOPEN ACCESS

Citation: Abrahams N, Mhlongo S, Chirwa E, Dekel B, Ketelo A, Lombard C, et al. (2024) Femicide, intimate partner femicide, and non-intimate partner femicide in South Africa: An analysis of 3 national surveys, 1999–2017. PLoS Med 21(1): e1004330. https://doi.org/10.1371/journal.pmed.1004330

Academic Editor: Charlotte Hanlon, Addis Ababa University / King's College London, ETHIOPIA

Received: June 26, 2023

Accepted: December 6, 2023

Published: January 18, 2024

Copyright: © 2024 Abrahams et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Access to a deidentified data is available upon reasonable request. A period of 24 months after publication of the main study results should elapse before requests are made, to allow the authors to publish sub-studies and further analyses. Request for the data must be made to hrec@samrc.ac.za.

Funding: The 2017 study was funded by - Ford Foundation (NA) Grant number 133505 <u>https://</u> www.fordfoundation.org/ - South African Medical Research Council (NA) <u>https://www.samrc.ac.za/</u> **RESEARCH ARTICLE**

Femicide, intimate partner femicide, and nonintimate partner femicide in South Africa: An analysis of 3 national surveys, 1999–2017

Naeemah Abrahams^{1,2*}, Shibe Mhlongo¹, Esnat Chirwa^{1,3}, Bianca Dekel^{1,4}, Asiphe Ketelo¹, Carl Lombard^{5,6}, Nwabisa Shai¹, Leane Ramsoomar¹, Shanaaz Mathews⁷, Gérard Labuschagne⁸, Richard Matzopoulos⁹, Megan Prinsloo⁹, Lorna J. Martin¹⁰, Rachel Jewkes^{1,4}

Gender & Health Research Unit, South African Medical Research Council, Cape Town, South Africa,
School of Public Health and Family Medicine: Faculty of Health Sciences, University of Cape Town, Cape Town, South Africa, 3 School of Public Health, University of Witwatersrand, Johannesburg, South Africa,
Office of the Executive Scientist, South African Medical Research Council, Pretoria, South Africa,
Biostatistics Unit, South African Medical Research Council, Pretoria, South Africa,
Biostatistics Unit, South African Medical Research Council, Cape Town, South Africa,
Biostatistics, Department of Global Health, Faculty of Medicine and Health Sciences, Stellenbosch University, Cape Town, South Africa, 7 Children's Institute, University of Cape Town, Cape Town, South Africa, 9 Burden of Disease Research Unit, South African Medical Research Council, Cape Town, South Africa, 10 Forensic Medicine & Toxicology, Department of Pathology, University of Cape Town, Cape Town, Cape Town, South Africa

* Naeemah.Abrahams@mrc.ac.za

Abstract

Background

In most countries, reliable national statistics on femicide, intimate partner femicide (IPF), and non-intimate partner femicide (NIPF) are not available. Surveys are required to collect robust data on this most extreme consequence of intimate partner violence (IPV). We analysed 3 national surveys to compare femicide, IPF, and NIPF from 1999 to 2017 using age-standardised rates (ASRs) and incidence rate ratios (IRRs).

Methods and findings

We conducted 3 national mortuary-based retrospective surveys using weighted cluster designs from proportionate random samples of medicolegal laboratories. We included females 14 years and older who were identified as having been murdered in South Africa in 1999 (n = 3,793), 2009 (n = 2,363), and 2017 (n = 2,407). Further information on the murdered cases were collected from crime dockets during interviews with police investigating officers. Our findings show that South Africa had an IPF rate of 4.9/100,000 female population in 2017. All forms of femicide among women 14 years and older declined from 1999 to 2017. For IPF, the ASR was 9.5/100,000 in 1999. Between 1999 and 2009, the decline for NIPF was greater than for IPF (IRR for NIPF 0.47 (95% confidence interval (CI) 0.42 to 0.53) compared to IRR for IPF 0.69 (95% CI 0.63 to 0.77). Rates declined from 2009 to 2017 and did not differ by femicide type. The decline in IPF was initially larger for women

The funders had no role in the study design, data collection, analysis, interpretation of the results, writing of the manuscript and decision to submit paper for publication.

Competing interests: The authors have declared that no competing interest interest exists.

Abbreviations: ASR, age-standardised rate; CAS, Crime Administration System; CI, confidence interval; COVID-19, Coronavirus Disease 2019; GBV, gender-based violence; GBVF, gender-based violence and femicide; ICC, intraclass correlation coefficient; IPF, intimate partner femicide; IPV, intimate partner violence; IRR, incidence rate ratio; MLM, medicolegal mortuary; NIPF, non-intimate partner femicide; SAPS, South African Police Services; UN, United Nations; UNODC, UN Office of Drugs and Crime; WHO, World Health Organization. aged 14 to 29, and after 2009, it was more pronounced for those aged 30 to 44 years. Study limitations include missing data from the police and having to use imputation to account for missing perpetrator data.

Conclusions

In this study, we observed a reduction in femicide overall and different patterns of change in IPF compared to NIPF. The explanation for the reductions may be due to social and policy interventions aimed at reducing IPV overall, coupled with increased social and economic stability. Our study shows that gender-based violence is preventable even in high-prevalence settings, and evidence-based prevention efforts must be intensified globally. We also show the value of dedicated surveys in the absence of functional information systems.

Author summary

Why was the study done?

- Femicide, the most severe form of gender-based violence, is not well described in middle- and low-income countries.
- Reliable routine data from police, justice and statistical departments are not available, and dedicated studies are needed to understand the scope of femicide in the country.

What did the researchers do and find?

- South Africa has a methodogy to collect national data from forensic and police services to estimate femicide and repeated surveys on 3 occasions (1999, 2009, and 2017) allowed for the assessement of change over 18 years.
- We found that an estimated 3,793 women 14 years and older were victims of femicide in 1999, and this decreased to an estimated 2,407 femicide victims in 2017. We further found a decrease in intimate partner femicide with the age-standardised rates decreasing from 9.5/100,000 population in 1999 to 4.9/100,000 population in 2017.

What do the findings mean?

- The 3 surveys over 18 years shows that femicide is preventable but that South Africa remains a country with the highest globally recorded rate of femicide.
- The reduction in femicide rates may be partially due to the contribution of activism of women and community-based organisations in bringing about change in Government's gender-based violence policy and programme.
- A concern is the increase in missing data from police sources over the years, which points to the urgent need to improve information systems and integration of data across departments.

Introduction

Femicide has been a relatively neglected area within the overall field of violence against women and girls. It was addressed for the first time at a global level in a 2013 resolution of the United Nations (UN) General Assembly "taking action against the gender-related killings of women and girls" (Resolution 68/191). A systematic review in the same year found that national data on the prevalence of IPF were only available from 66 countries, and these were mostly high-income. It concluded that 13.5% of all homicides were committed by intimate partners, with women 6 times more likely affected than men [1]. A recent report of the UN Office of Drugs and Crime (UNODC) used the term "femicide" to refer to the intentional killing of women with a gender-related motivation and clarified this as referring to murders by an intimate partner or a family member [2]. It concluded that 58% of all female murders globally were perpetrated by an intimate partner or a family member, a rate of 1.1/100,000 female population, and the highest rates were reported in Surinam (4.3/100,000 female population) and Belize (2.1/100,000 female population) [2]. In South Africa, information on IPF is not available from routine statistics such as from police, justice, and home affairs death registration databases [3,4]. However, in 2009, the rate measured in a national survey was 5.6/100,000, which was 5 times higher than the UNODC-estimated overall global rate [5].

This paper reports the findings of 3 national surveys on femicide in South Africa conducted across 18 years. We use the term "femicide" to refer to the intentional killing of women and girls. This includes intimate partner femicide (IPF), refering to the murder of a woman by a current or former intimate partner (e.g., husband or boyfriend), a definition aligned to that of intimate partner homicide used by, for example, the Office of Justice in the United States of America and the Australian Bureau of Statistics, as well as non-intimate partner femicide (NIPF) (i.e., killing by someone else) [6,7]. This differs from the UNODC definition above, and indeed definitional differences are a challenge for global comparisons [6]. The 2021 UNODC report mainly used countries' statistical data but noted that, in most lower- and mid-dle-income countries (LMICs), police administrative data are not a reliable source of information on IPF because perpetrator information is frequently not recorded; thus, the relationship between the victim and the perpetrator is not known [8].

In South Africa, the solution to the problem of data access has been circumvented by the development of a methodology for conducting national cluster sample surveys, and repetition of these surveys at intervals enabled femicide surveillance. This paper presents the findings of 3 surveys over 18 years and compares femicide, IPF, and NIPF between 1999 and 2009 and between 2009 and 2017 in South Africa.

Methods

Study background and data sources

In South Africa, "unlawful and intentional killing" is referred to as murder. We conducted 3 national retrospective studies of the murder of female victims aged 14 years and older, using a common approach in all respects with some variations (see details below). Our primary interest was to understand IPF and choose the starting age of 14 years as this is commonly the age when South African young women start dating. The definitions used in the study are shown in Box 1.

Our 2 main categories were IPF, e.g., women killed by a current or ex-intimate partner, and NIPF, included women killed by everyone else other than an intimate partner, which include family members as perpetrators. Two data sources were used in this study. The primary data source was medicolegal mortuary (MLM) records of murdered women, and the second source

Box 1. Femicide definitions used in the 3 South African Femicide Surveys

Femicide-Murder of a woman

Intimate Partner Femicide (IPF)–Murder of a woman by an intimate partner (i.e., a current or ex-husband/boyfriend, same-sex partner, or a rejected would-be lover)

Non-Intimate Partner Femicide (NIPF)–Murder of a woman by someone other than an intimate partner (i.e., stranger, family member, acquaintance, etc.)

was data from police dockets based on the police investigation of each murdered women in the sample. The primary data source of MLM data was used because according to the Inquests Act of 1959, all cases of unnatural deaths must undergo a postmortem examination and records of all unnatural deaths, which include murder cases, are available from MLMs [9].

Study design, sample, and sampling process

To draw national representative samples for each survey, we developed a sampling frame listing all MLMs operating in the country in the study year. The MLM was stratified by size based on the number of autopsies performed per annum, i.e., small MLM = <500 autopsies, medium size MLM = 500 to 1,499 autopsies performed per annum, and large MLM = >1,499 autopsies per year as shown in Table 1. The sample selection for all 3 surveys was done by the study statistician (CL).

The sampling frame was prepared in an MS Excel worksheet, and the relevant stratification was applied. Due to the stratification by size, simple random sampling within strata was used for selection of mortuaries using the uniform random number generator available in MS Excel. To avoid single-unit strata, a minimum number of the mortuaries per stratum was selected (n = 2). The stratification ensured that the sample was representative of both small rural mortuaries and larger ones attached to medical schools, and enhanced precision of national estimates. The number of MLM differed across the 3 studies. In the first study (1999), we based our assumption about the sample size on available surveillance data and a small study conducted in Gauteng [10].

The differences in the sampling frame was due to rationalisation of MLM after 1999 and ongoing operational changes to the forensic services, it meant that the number of mortuaries and their sizes differed for each of the 3 study years. In addition, the larger sampling fraction for each successive year was due to better research funding. While the sample size for the 1999

Mortuary strata]	1999	2	2009	2017		
	Number of mortuaries	Sample (Sampling Fraction)	Number of mortuaries	Sample (Sampling Fraction)	Number of mortuaries	Sample (Sampling Fraction)	
Small: <500 autopsies performed per year	176	12 (6.8)	81	20 (24.7)	96	55 (57.3)	
Medium: 500–1,499 autopsies performed per year	34	5 (14.7)	33	13 (39.4)	31	19 (61.3)	
Large: >1,499 autopsies performed per year	15	8 (53.3)	8	5 (55.5)	10	7 (70.0)	
Total	225	25 (11.1)	122	38 (31.1)	137	81 (59.1)	

Table 1. Sampling frame and sampling fraction used across the three surveys.

https://doi.org/10.1371/journal.pmed.1004330.t001

and 2009 studies was designed to produce national estimates, the 2017 sample size was intended to produce provincial estimates.

Case identification and data collection

At the sampled mortuaries, we identified females (sex as recorded in the autopsy report) aged 14 and older who were registered between 1 January and 31 December in the year of study with death due to murder. We collected data from the MLM file and autopsy report for each case. Approval was granted by national and provincial police management who facilitated the identification of cases using the South African Police Services (SAPS) Crime Administration System (CAS) number documented in the MLM files to link each death with a police investigation docket. In 2017, we requested from the Research Ethics Committee to use personal identification number to assist in tracing murder cases in the police system if CAS numbers were absent. The research assistants sought consent from the police officers and, using a questionnaire, asked them questions about the case, which the police answered by extracting the information from the police dockets.

We excluded cases where both age and sex could not be determined and cause of death remained undetermined. Sex was unknown in 50 cases in the 2017 study, and these were usually cases which were decomposed bodies. We also excluded 9 cases where both age and sex were unknown. Autopsy reports include description of the deceased such as "adult" or "child," and we used this information if date of birth was unknown and included adult female cases. These cases with no date of birth but identified as "adults" were excluded from the age group analysis. We also excluded if the cause of death was remained undetermined. In 2017, we decided to do interviews on the undetermined cause of death cases. A total of 502 cases identified at the mortuaries had undetermined cause of death. These were deaths related to poisoning, fire injuries, or deaths where the forensic report could not identify a cause, e.g., a woman found on her bed with no evidence of force entry or injuries on body. These cases were investigated by the police as "inquest cases," and during police interviews, we found that 62 of the undetermined cases were excluded.

In 1999 and 2009, we used a paper-based questionnaire to collect data, and in 2017, we used REDCap, a web-based data entry tool [11]. At the MLMs, we collected data on age, cause of death (stab, gunshot, blunt force, etc.), CAS number, name of police station, samples taken, and if forensic evidence were collected. We also identified rape femicides (which could be in either and IPF or NIPF case) and collected data on whether rape was identified or suspected, i.e., a sexual assault evidence kit was used, victim's underwear removed, or genital injuries.

The first author, as Principal Investigator, trained research assistants in all 3 studies to collect data at the MLMs and police. Research assistants were always supported by a study coordinator. In all studies, interviews with the police were either conducted in-person or telephonically with case files (dockets) the primary source of data from which the police officers extracted the data. The study team was not allowed to handle police dockets. We sought to interview the primary investigating officer, and, if this was not possible, a senior detective extracted the information from the case docket. A single police officer may have been investigating more than one murder (if more than one murder located at a police station) and often provided data on more than one case. We asked whether murder was suspected, additional victim information (education, employment), and information on the perpetrator, which revealed the victim-perpetrator relationship and was crucial to identify the case as an IPF (perpetrator was an intimate partner) or an NIPF (perpetrator was a non-partner). We also collected data on perpetrator age, sex, education, employment, known use of alcohol, firearm ownership, and prior convictions. For IPF perpetrators, we collected data on any prior history of IPV. We also gathered further information on the crime, its investigation, and legal outcome. For cases that were not definitively closed, we followed convention in the field of research on homicides used in the supplementary homicide data by the Bureau of Justice in the US where the perpetrator is defined as the person the Investigating Officer perceives to be primarily responsible [12].

In 1999, we collected data approximately 3 years after the deaths (in 2002 to 2003) in order to allow time for the legal cases to have been closed. In both the 2009 and 2017, studies data collection started within 2 years after the deaths. The 2017 data collection extended over a longer period ending only in November 2021, due to lockdown restrictions during the Coronavirus Disease 2019 (COVID-19) pandemic.

Race remains an important factor for risk profiles for violence and health in South Africa, and we define racial categories following conventions that were developed under Apartheid, which may not have the same meaning as for other countries and regions. It is acknowledged that the notion of race is a social construct, and in South Africa, "coloured" race refer to a mixed but a culturally specific group.

Data management and statistical analysis

The protocol approved by the Ethics Committee include an analysis plan and is presented in the Supporting Information file (S1 Text). The analysis in this manuscript followed what was done in the analysis that compared 1999 and 2009 surveys [5]. The analysis related to imputing missing data on victim perpetrator relationships was not anticipated and not presented in the protocol analysis plan (S1 Text) and is described below. This change was necessitated by the large proportion of missing information due to the perpetrator not being identified by police. Data management and analysis for all the 3 studies were done in MS Excel and Stata 17. Procedures for cleaning and management were similar across the 3 studies. A difference was that a paper-based data collection process was followed in 1999 and 2009 studies, while a web-based data collection was followed in 2017. The PI (NA), senior statistician (CL), and CO PI (RJ) were involved in all 3 study procedures. In 2017, the statisticians (SM and EC) took more responsibilities for recoding, using the codes prepared in the 2 earlier studies.

Analysis weighting was based on the first 2 surveys designed to report on the national level only, and the strata were based on mortuary size. In 2017, additional funding allowed us to increase the sample size to allow for both national and provincial analysis. Province as a stratum allowed us to have femicide estimates at the provincial, which is critical for planning prevention and response to gender-based violence and femicide (GBVF) in the country as noted in the national strategic plan [13]. South African provinces are different across domains including population structures (age, education, etc.) and other socioeconomic factors including levels of crime [14,15]. Therefore, for the 2017 survey, stratification was done by mortuary size and province. The survey analysis weights were applied to account for the selection probabilities were based on the number of mortuaries randomly selected in each stratum of the total number of facilities.

Weight calculations were performed in Microsoft Excel using the formula:

Weight = realisation weight (RW) \times 1/sampling fraction (SF) \times 1/ primary sampling unit selection probability (PSUP)

where

RW = expected sample/realised sample for each mortuary

SF = 1/2 for mortuaries that had half their postmortem folders surveyed

SF = 1 for mortuaries that had all their postmortem folders surveyed

PSUP = <u>number of mortuaries selected for stratum</u>

total number of mortuaries in stratum

For analysis, we considered the 3 studies to be independent surveys due to the time separation and the independent sampling procedures used for each. All analysis procedures considered the multistage design of the study as described above with weighting stratified by mortuary size (and in 2017, also by province) and the mortuaries as clusters. After defining the datasets as survey data using "svyset," "svy" command was used to account for stratification and weights. Taylor linearization was used to get robust variance estimators for parameter estimates.

To calculate the age-standardised rates (ASRs), we used the female population estimates derived from a mathematical model (Thembisa model version 4.4) for demographic statistics. This is a source used extensively for all government and adminstrative purposes at national and provincial levels [16,17]. We also used the World Health Organization (WHO) world standard population distribution normalised weightings for the ASR calculation to take into account the subgroup analysis for the ages 14 and above and for comparison with other countries [18]. The rates were calculated to account for missing ages (see details in Table 2 footnotes). To calculate rates by race, we used the midyear population estimates from Statistics South Africa [19].

We presented overall femicide statistics and those for IPF and NIPF. To account for missing information due to the perpetrator not being identified by police, we used the Hot deck multiple imputation procedure to allocate cases to the IPF and NIPF groups [20]. The Hot deck imputation is a commonly used imputation method for surveys. It allowed for the multilevel structure of the data to be taken into account to maintain the clustering across the mortuaries. If this is ignored, the clustering is diluted, and, hence, the effective sample size is inflated. Thus, imputation was carried out within the province and mortuary size strata to preserve the study design. Resampling was straightforward in the large and medium sized mortuaries, but not in the small mortuaries. For the latter, we considered the resampling from the pooled province by place-ofinjury data. Place of injury was included as a theoretically known variable that predicts intimate femicide and non-intimate femicide. It is known that intimate femicide deaths commonly occur in a home setting. The inclusion of place of injury in the imputation procedure was done to improve on the classification of unknown cases as intimate or non-intimate femicide cases. We compared the intraclass correlation coefficient (ICC), which is an indicator of the extent of clustering. Using standard univariate logistic regression multiple imputation, the ICC become much smaller, whereas the multiple Hot deck imputation maintained the ICC since it took multilevel design into account. This procedure was followed in all the 3 surveys, and 30 Hot deck imputations were used in each survey. In each survey, the 30 Hot deck multiple imputed datasets were then used to get pooled parameter estimates and their standard errors. Although Hot deck imputation has a limitation on the number of variables that can be included in the procedure and may not perform very well for large proportions of missing data as was the case for the 2017 survey, the multiple imputation and inclusion of a predictor of intimate femicide helped in improving the precision of the estimates.

We summarised all data, both imputed and non-imputed, using weighted counts and percentages and estimated the homicide ASRs for all female homicides in 1999, 2009, and 2017 and within femicide subgroups (IPF and NIPF). We calculated incidence rate ratios (IRRs) and their respective 95% confidence intervals (CIs) to compare rates between 2009 and 1999, and between 2017 and 2009. The standard errors for the rates and IRRs took into account the design effects of each survey.

This study is reported as per the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) checklist for cross sectional studies (see <u>S1 STROBE</u> Checklist).

Table 2. Age-standardised population rates for 1999, 2009, and 2017 for all female murders: IPF and NIPF by age and IRRs of population rate estimates between study years: weighted and imputed data.

	1999 Unweighted = 1,052 Weighted = 3,793			2009 Unweighted = 930 Weighted = 2,363			IRR of population rate estimates: 2009/1999		2017 Unweighted = Weighted = 2	IRR of population rate estimates: 2017/2009	
	N	Percent (95% CI)	Rate per 100,000 pop (95% CI)	N	Percent (95% CI)	Rate per 100,000 pop (95% CI)	(95% CI)	N	Percent (95% CI)	Rate per 100,000 pop (95% CI)	(95% CI)
Overall female murders*	3,793		24.2 (15.5–32.9)	2,363		12.6 (8.5–16.6)	0.52 (0.48-0.56)	2,407		11.1 (9.8–12.4)	0.88 (0.81–0.95)
IPF	1,610	48.9 (42.0-55.7)	9.5 (6.4–12.7)	1,294	57.1 (52.7-61.6)	6.6 (5.3-8.0)	0.69 (0.63-0.77)	1,089	55.0 (51.6-58.4)	4.9 (4.1-5.8)	0.74 (0.66-0.83)
NIPF	1,686	51.1 (44.3-58.0)	11.4 (6.9–15.9)	971	42.9 (38.4-47.3)	5.4 (4.2-6.6)	0.47 (0.42-0.53)	890	45.0 (41.6-48.4)	4.2 (3.4-4.9)	0.78 (0.69-0.88)
Overall femicide by age group*											
14-29 y	1,359	35.8 (31.7-40.2)	6.8 (4.7-8.9)	867	36.7 (33.8–39.6)	3.8 (2.6-5.0)	0.56 (0.50-0.63)	943	39.2 (37.7-40.7)	4.2 (3.8-4.6)	1.11 (0.97–1.26)
30-44 y	1,310	34.5 (29.8-39.6)	8.1 (5.7–10.5)	784	33.2 (30.0-36.5)	4.2 (2.9–5.6)	0.52 (0.46-0.59)	774	32.2 (30.2-34.2)	3.4 (2.9–3.8)	0.81 (0.71-0.93)
45-59 y	512	13.5 (10.9–16.6)	4.4 (3.0-5.9)	408	17.3 (14.9–19.9)	2.5 (1.7-3.3)	0.57 (0.47-0.68)	365	15.2 (13.0–17.6)	1.8 (1.6–2.1)	0.72 (0.59-0.88)
60+ y	391	10.3 (6.4–16.4)	3.4 (1.2–5.7)	290	12.3 (9.9–15.1)	2.0 (1.3-2.7)	0.59 (0.48-0.73)	303	12.6 (11.2–14.2)	1.6 (1.4–1.8)	0.80 (0.64-1.00)
Undetermined age	222	5.8 (3.6-9.3)		15	0.6 (0.3–1.3)			22	0.3 (0.1–0.6)		
IPF by age group											
14-29 y	739	45.9 (38.3-53.4)	3.7 (2.7-4.7)	540	41.8 (37.1-46.4)	2.4 (1.9-2.8)	0.65 (0.56-0.76)	446	40.9 (37.7-44.2)	2.0(1.7-2.3)	0.95 (0.80-1.14)
30-44 y	579	35.9 (28.2-43.7)	3.6 (2.6-4.6)	534	41.3 (36.4-46.1)	2.9 (2.4-3.4)	0.81 (0.68-0.95)	417	38.3 (35.3-41.3)	1.8 (1.6-2.1)	0.78 (0.65-0.94)
45-59 y	127	7.9 (5.3–10.6)	1.1 (0.6–1.6)	156	12.1 (9.3–14.9)	1.0 (0.7–1.2)	0.91 (0.66-1.26)	137	12.5 (9.6-15.5)	0.7 (0.5-0.8)	1.00 (0.69-1.45)
60+ y	63	3.9 (1.1-6.7)	0.5 (0.1-1.0)	59	4.6 (2.6-6.5)	0.4 (0.2-0.6)	0.80 (0.49-1.31)	76	7.0 (4.9-9.0)	0.4 (0.3-0.5)	2.00 (0.84-4.75)
Undetermined age	102	6.4 (1.7–11.0)		4	0.3 (0.0-0.8)			14	1.3 (0.3–2.3)		
NIPF by age groups											
14-29 y	477	28.3 (22.2-34.4)	2.4 (1.5-3.2)	299	30.8 (26.9-34.6)	1.3 (1.0–1.6)	0.54 (0.44-0.66)	319	35.9 (32.4-39.3)	1.4 (1.2–1.6)	1.08 (0.86-1.34)
30-44 y	546	32.4 (26.1-38.7)	3.4 (2.4-4.4)	234	24.0 (20.0-28.0)	1.3 (1.0–1.5)	0.38 (0.31-0.47)	243	27.3 (22.9–31.7)	1.1 (0.8–1.3)	0.85 (0.66-1.09)
45-59 y	323	19.2 (14.4–23.9)	2.8 (1.7-3.9)	217	22.4 (18.4-26.4)	1.3 (1.0–1.7)	0.46 (0.37-0.59)	149	16.7 (13.4–20.1)	0.7 (0.6-0.9)	0.54 (0.40-0.72)
60+ y	271	16.1 (8.8-23.4)	2.4 (1.0-3.7)	222	22.8 (17.9–27.7)	1.5 (1.1–1.9)	0.63 (0.49-0.80)	171	19.2 (16.1–22.4)	0.9 (0.7-1.1)	0.60 (0.45-0.79)
Undetermined Age	69	4.1 (0.5–7.6)						8	0.9 (0.0–1.9)		
Overall femicide by race group*											
African	3,019	80.3 (67.0-89.1)	25.8 (14.4-37.3)	1,884	80.3 (69.1-88.1)	12.9 (9.2–16.5)	0.50 (0.46-0.54)	2,078	86.4 (83.9-88.5)	12.3 (11.2–13.5)	0.96 (0.88-1.05)
Coloured	516	13.7 (6.3–27.5)	37.5 (6.7-68.3)	309	13.2 (6.2–25.9)	17.7 (3.4–32.0)	0.47 (0.39-0.57)	233	9.7 (7.9–11.9)	11.9 (9.5–14.4)	0.67 (0.53-0.85)
Indian	41	1.1 (0.4–3.0)	9.7 (0.0–19.9)	31	1.3 (0.5–3.7)	5.9 (0.0-12.4)	0.61 (0.32–1.16)	21	0.9 (0.6–1.4)	3.6 (2.0-5.2)	0.60 (0.28-1.30)
White	183	4.9 (2.9-8.2)	9.3 (2.9–15.7)	116	4.9 (2.8-8.5)	5.5 (2.0-9.1)	0.60 (0.43-0.82)	72	3.0 (2.4–3.7)	3.4 (2.6-4.3)	0.62 (0.42-0.94)
IPF by race group											
African	1,255	78.2 (66.7-89.7)	10.7 (7.7–13.8)	1,041	80.6 (72.3-88.9)	7.1 (6.0–8.2)	0.66 (0.59–0.74)	948	86.5 (83.6-89.4)	5.6 (4.9-6.3)	0.79 (0.70-0.89)
Coloured	271	16.9 (5.7-28.1)	19.7 (5.6-33.7)	179	13.9 (5.1–22.6)	10.3 (3.4–17.2)	0.52 (0.40-0.68)	115	10.5 (7.7–13.3)	5.9 (4.2-7.5)	0.57 (0.41-0.79)
Indian	22	1.4 (0.2–2.6)	5.2 (1.0-9.4)	21	1.6 (0.4–2.8)	4.0 (1.1-6.9)	0.76 (0.33-1.76)	9	0.8 (0.1–1.6)	1.5 (0.1–2.9)	0.38 (0.13-1.13)
White	57	3.5 (1.9-5.2)	2.9 (1.4-4.4)	48	3.7 (1.8-5.7)	2.3 (1.1-3.5)	0.79 (0.46 1.35)	24	2.2 (1.3-3.1)	1.1 (0.7–1.6)	0.50 (0.25-0.99)
NIPF by race groups											
African	1,296	78.2 (66.4–90.0)	11.1 (7.3–14.9)	760	79.0 (70.2-87.9)	5.2 (4.2-6.2)	0.47 (0.41-0.53)	788	87.4 (84.8-89.9)	4.7 (4.2–5.2)	0.90 (0.78-1.03)
Coloured	225	13.6 (3.2–24.0)	16.4 (4.1-28.6)	126	13.1 (5.0–21.3)	7.2 (2.5–12.0)	0.44 (0.33-0.60)	83	9.2 (7.2–11.1)	4.2 (3.4–5.0)	0.59 (0.40-0.86)
Indian	15	0.9 (0.3-1.6)	3.6 (1.2-6.0)	10	1.1 (0.0-2.2)	1.9 (0.0-4.0)	0.53 (0.18-1.62)	0			

(Continued)

Table 2. (Continued)

	1999 Unweighted = 1,052 Weighted = 3,793			2009 Unweighted = 930 Weighted = 2,363			IRR of population rate estimates: 2009/1999		2017 Unweighted = Weighted = 2	1,301 ,407	IRR of population rate estimates: 2017/2009
	N	Percent (95% CI)	Rate per 100,000 pop (95% CI)	N	Percent (95% CI)	Rate per 100,000 pop (95% CI)	(95% CI)	N	Percent (95% CI)	Rate per 100,000 pop (95% CI)	(95% CI)
White	121	7.3 (3.9–10.6)	6.1 (2.8-9.4)	63	6.6 (3.2-10.0)	3.0 (1.5-4.5)	0.49 (0.32-0.75)	31	3.5 (2.3-4.6)	1.5 (1.0-2.0)	0.50 (0.27-0.90)

All analysis on ASRs was calculated using the Thembisa model except for race group that used Statistics South Africa midpopulation estimates. **1999 female population**: Overall: 15,775,803; By Age: 14–29 years: 6,976,810; 30–44 years: 4,564,389; 45–59 years: 2,427,045; and 60 + years: 1,807,559. **2009 female population**: Overall: 18,982,433; By Age: 14–29 years: 7,960,305; 30–44 years: 5,248,792; 45–59 years: 3,460,820; and 60 + years: 2,312,516. **2017 female population**: Overall: 21,520,499; By Age: 14–29 years: 7,872,159; 30–44 years: 6,459,190; 45–59 years: 4,195,543; and 60 + years: 2,993,607. For race groups age specific rates are reported using Statistics South Africa mid population estimates:

1999 female population: Overall: 15,458,162; By race: African: 11,683,651; Coloured: 1,375,413; Indian: 424,331; and White: 1,984,767.

2009 female population: Overall: 19,027,717; By race: African: 14,655,388; Coloured: 1,744,521; Indian: 529,557; and White: 2,098,251.

2017 female population: Overall: 21,501,234; By race: African: 16,862,856; Coloured: 1,958,404; Indian: 595,297; and White: 2,084,677.

ASRs were calculated to accounted for undetermined age by multiplying the ASR not accounting for undetermined age with a factor calculated by taking total number of female homicide cases divided by the total number of female homicide cases minus the total number of female homicide cases with undetermined age.

IRR is calculated by taking the fractions of the exposed and their total population. The 95% CIs are then adjusted from normal approximation calculation to account for the design effect of the survey.

* No imputation on overall descriptive summaries.

ASR, age-standardised rate; CI, confidence interval; IPF, intimate partner femicide; IRR, incidence rate ratio; NIPF, non-intimate partner femicide.

https://doi.org/10.1371/journal.pmed.1004330.t002

Ethical approval

Ethical approval for the study was granted by the Ethics Committee of the South African Medical Research Council (EC 008-5-2018), and further approval and access to data were obtained from the National and Provincial Departments of Health, the Forensic Pathology Service, and the South African Police Service. The study was based on people (the victims) who have died, and consent was therefore not required. Also, data for both victims and perpetrators were analysed anonymously.

Results

Weighted estimates and missing data

The weighted estimates of the number of murdered women 14 years and older across the 3 studies are presented in Fig 1. All sampled mortuaries contributed data. The greatest proportion of cases with police data not traced in the police system or police interviews not done was found in 2017, with no police interviews done for 15.7% of the victims identified at the mortuaries. We, however, found similar proportion of missing data from police across the 3 strata of mortuaries (see <u>S1</u> Table). Among the cases with police interviews done and investigation data identified, we found a successive increase in the proportion of perpetrators not identified by the police during the investigation, from 18.6% of cases in 1999 to 30% in 2017, hence the need for imputation.

Age-standardised rates (ASRs) and incident rate ratios (IRRs)

Table 2 shows the age-standardised population rates (ASRs) for the different subgroups of femicide, as well as the IRRs of the population estimates comparing 1999 with 2009 and 2009 with 2017. In 2017, the ASR for overall femicide was 11.1/100,000 (95%CI 9.8 to 12.4) female population, the ASR for IPF was 4.9/100,000 (95% CI 4.1 to 5.8) female population, and that for NIPF was 4.2/100,000 (95% CI 3.4 to 4.9) female population. The ASRs for all femicides



Fig 1. Flow chart of all estimated femicide cases of women 14 years and older across the 3 studies: weighted and imputed data.

https://doi.org/10.1371/journal.pmed.1004330.g001

declined from 24.2/100,000 (95% CI 15.5 to 32.9) female population in 1999 to 12.6/100,000 (95% CI 8.5 to 16.6) female population in 2009. A similar reduction in IPF was found from 9.5/100,000 (95% CI 6.4 to 12.7) in 1999 to 6.6/100,000 (95% CI 5.3 to 8.0) in 2009. In 2017, the rate for IPF was 4.9/100,000 (95% CI 4.1 to 5.8) female population. A similar decrease was found in the NIPF rates, which were 11.4/100,000 (95% CI 6.9 to 15.9) in 1999, 5.4/100,000 (95% CI 4.2 to 6.6) in 2009, and 4.2/100,000 (95% CI 3.4 to 4.9) in 2017. The IRRs of the population estimates showed that the declines between 1999 and 2009 and 2009 and 2017 were statistically significant for all femicide types. However, it was notable that between 1999 and 2009, the decline for NIPF was greater than for IPF (IRR for NIPF 0.47 (95% CI 0.42 to 0.53)) compared to IRR for IPF 0.69 (95% CI 0.63 to 0.77). A similar difference by femicide type was not seen between 2009 and 2017 (IRR for IPF = 0.74 (95% CI 0.66 to 0.83)) versus IRR for NIPF = 0.78 (95% CI 0.69 to 0.88)).

Age-standardised rates and incident rate ratios by age and race

In all the surveys, the IPF ASRs by age groups showed higher rates for the age groups 14 to 29 and 30 to 44 than for the age groups from 45 years and over. The decline in IPF from 1999 to 2009 was greater for younger women (aged 14 to 44 years) than older women (45 years old and older). A greater reduction in IPF was seen for the 14- to 29-year age group (IRR 0.69 (95% CI 0.63 to 0.77)) than for the 30- to 44-year group (IRR 0.81 (95% CI 0.68 to 0.95)) from 1999 to 2009. Between 2009 and 2017, the decline in IPF by age group was only significant for women aged 30 to 44 years (IRR 0.78 (95% CI 0.65 to 0.94)). There was a decline in NIPF between 1999 and 2009 across all age groups, and between 2009 and 2017, this was seen for the

age groups of women aged 45 and over. The same analysis was done on the non-imputed data, and a similar pattern of decrease was shown (<u>S2 Table</u>).

Provincial estimates and rates

We present the provincial estimates and rates of overall femicide, IPF, and NIPF from the 2017 study in Table 3. The Eastern Cape Province had the highest overall femicide rate at 22.3/ 100,000 female population, which was nearly double the province with the next highest rate, i.e., KwaZulu-Natal Province at 14.0/100,000 female population. The lowest rate was found in the Limpopo Province at 4.9/100,000 female population. The Eastern Cape Province also had the highest rates of both IPF and NIPF.

Table 3. Age-standardised population rates for 2017 for all female murders, IPF, and NIPF by South African Provinces: weighted and imputed data.

Provinces	Female Population 14 years+	Overall Female Murders						
		N (95% CI)	% (95% CI)	Rate/100,000 population (95% CI)				
Western Cape	2,575,176	316 (309–323)	13.1 (12.0–14.3)	12.3 (12.0–12.5)				
Eastern Cape	2,533,854	565 (506–624)	23.5 (20.9–26.3)	22.3 (20.0-24.6)				
Northern Cape	389,291	43 (21–65)	1.8 (1.1–3.0)	11.1 (5.3–16.8)				
Free-State	1,060,513	137 (117–158)	5.7 (4.8-6.7)	12.9 (11–14.9)				
Kwa-Zulu Natal	4,165,842	583 (490–675)	24.2 (20.9–27.8)	14.0 (11.8–16.2)				
North West	1,358,271	105 (97–113)	4.4 (4.1–4.7)	7.7 (7.1–8.3)				
Gauteng	5,630,018	458 (299–617)	19.0 (14.2–25.0)	8.1 (5.3–11)				
Mpumalanga	1,687,111	96 (69–123)	4.0 (3.0-5.3)	5.7 (4.1-7.3)				
Limpopo	2,105,317	104 (84–124)	4.3 (3.5–5.3)	4.9 (4.0-5.9)				
Overall	21,520,499	2,407 (2,204–2,610)	100.0	11.2 (10.2–12.1)				
		IPF						
Western Cape	2,575,176	125 (115–135)	49.9 (28.6–71.2)	4.9 (4.5–5.3)				
Eastern Cape	2,533,854	203 (155–252)	44.7 (32.6–56.8)	8 (6.1–9.9)				
Northern Cape	389,291	24 (11–38)	68.1 (25.7–100.0)	6.3 (2.8–9.7)				
Free-State	1,060,513	62 (52–71)	54.3 (40.6-67.9)	5.8 (4.9–6.7)				
Kwa-Zulu Natal	4,165,842	246 (190–302)	52.4 (44.4-60.3)	5.9 (4.6-7.3)				
North West	1,358,271	50 (40-59)	58.5 (17.8–99.2)	3.7 (3-4.4)				
Gauteng	5,630,018	216 (133–299)	58.1 (23.2-93.1)	3.8 (2.4–5.3)				
Mpumalanga	1,687,111	53 (31–74)	62.5 (46.6–78.5)	3.1 (1.9–4.4)				
Limpopo	2,105,317	49 (29–70)	49.1 (7.7–90.6)	2.3 (1.4–3.3)				
Overall	21,520,499	1,029 (911–1,146)	52.3 (49.1-55.5)	4.8 (4.2-5.3)				
		NIPF						
Western Cape	2,575,176	126 (116–136)	50.1 (28.8-71.4)	4.9 (4.5–5.3)				
Eastern Cape	2,533,854	252 (217–286)	55.3 (43.2-67.4)	9.9 (8.6–11.3)				
Northern Cape	389,291	11 (6–17)	31.9 (0.0–74.3)	2.9 (1.5-4.4)				
Free-State	1,060,513	52 (40-65)	45.7 (32.1–59.4)	4.9 (3.8–6.1)				
Kwa-Zulu Natal	4,165,842	224 (175–273)	47.6 (39.7–55.6)	5.4 (4.2–6.6)				
North West	1,358,271	35 (28-42)	41.5 (0.8-82.2)	2.6 (2.1–3.1)				
Gauteng	5,630,018	156 (119–193)	41.9 (6.9–76.8)	2.8 (2.1–3.4)				
Mpumalanga	1,687,111	31 (22–41)	37.5 (21.5–53.4)	1.9 (1.3–2.4)				
Limpopo	2,105,317	51 (32–70)	50.9 (9.4-92.3)	2.4 (1.5–3.3)				
Overall	21,520,499	938 (862-1015)	47.7 (44.5-50.9)	4.4 (4.0-4.7)				

CI, confidence interval; IPF, intimate partner femicide; NIPF, non-intimate partner femicide.

https://doi.org/10.1371/journal.pmed.1004330.t003

Mechanisms of death

We present trends in the 3 main mechanisms of death in <u>Table 4</u>. Death by gunshot showed a decrease for all forms of femicide between 1999 and 2009; however, this did not continue between 2009 and 2017 for NIPF. Similarly, there was a decrease in rates of death by stabbing for all forms of femicide between 1999 and 2009; however, an increase was found for all femicides between 2009 and 2017 (IRR 1.17 (95% CI 1.02 to 1.35)). Rates of death from blunt force injuries showed a continued decline with a difference found for all forms of femicide. Overall, a higher proportion of IPF victims died from blunt force injuries compared to the NIPF group. The analysis on the non-imputed data shows similar patterns of distribution and decreases (S3 Table).

Table 4. Age-standardised population rates for 1999, 2009, and 2017 for all female murders, IPF, and NIPF by the mechanism of death and IRRs of population rate estimates between study years: weighted and imputed data.

Characteristics 1999				2009		IRR of population	2017			IRR of population	
	Overall Unweighted = 1,052 Overall Weighted = 3,793 IPF Weighted = 1,610 NIPF Weighted = 1,686			Overall Unweighted = 930 Overall Weighted = 2,363 IPF Weighted = 1,294 NIPF Weighted = 971		rate estimates: <u>2009/1999</u> (95% CI)	Ove Ove]	erall Unweigh verall Weight IPF Weighted NIPF Weight	nted = 1,301 eed = 2,407 d = 1,089 eed = 890	rate estimates: <u>2017/2009</u> (95% CI)	
	N	Percent (95% CI)	Rate per 100,000 pop (95% CI)	N	Percent (95% CI)	Rate per 100,000 pop (95% CI)		N	Percent (95% CI)	Rate per 100,000 pop (95% CI)	
Firearm deaths											
All female murders*	1,147	33.4 (24.9– 43.1)	7.2 (2.7– 11.7)	462	19.5 (15.1– 25.0)	2.4 (1.2–3.6)	0.33 (0.29–0.39)	563	23.4 (22.2– 24.7)	2.6 (2.1–3.1)	1.08 (0.91–1.29)
IPF	495	31.9 (22.1– 41.8)	2.9 (1.4-4.5)	259	20.0 (14.9– 25.1)	1.3 (0.8–1.9)	0.45 (0.36-0.55)	215	19.8 (16.6– 22.9)	1.0 (0.6–1.4)	0.77 (0.60–0.99)
NIPF	556	34.7 (24.8– 44.5)	3.8 (1.5-6.0)	190	19.6 (14.6– 24.6)	1.0 (0.5–1.5)	0.26 (0.21-0.33)	213	24.0 (21.0– 26.9)	1.0 (0.7–1.3)	1.00 (0.76–1.31)
Stab injury deaths											
All female murders*	1,049 [¥]	30.5 (24.0- 37.9)	6.6 (3.1– 10.0)	668	28.3 (23.6– 33.5)	3.5 (2.0–5.0)	0.53 (0.46-0.61)	897 [¥] ¥	37.6 (35.8– 39.4)	4.1 (3.4–4.8)	1.17 (1.02–1.35)
IPF	468	30.2 (23.7– 36.7)	2.8 (1.3-4.4)	356	27.5 (22.7– 32.3)	1.8 (1.2–2.4)	0.64 (0.53–0.78)	423	38.9 (35.5- 42.3)	1.9 (1.4–2.4)	1.06 (0.87–1.28)
NIPF	509	31.8 (23.0- 40.6)	3.3 (1.6-5.1)	301	31.0 (25.7– 36.4)	1.6 (1.1–2.2)	0.48 (0.40-0.59)	331	37.2 (33.5– 40.9)	1.5 (1.1–2.0)	0.94 (0.75–1.16)
Blunt force injury deaths											
All female murders*	943	24.9 (17.7– 33.8)	6.0 (2.5–9.6)	580	24.5 (20.5– 29.0)	3.1 (1.7-4.5)	0.52 (0.45-0.60)	501	20.8 (19.3– 22.3)	2.3 (1.9–2.8)	0.74 (0.63-0.88)
IPF	515	32.0 (21.5– 42.5)	3.0 (1.4-4.7)	354	27.4 (23.0– 31.7)	1.8 (1.1–2.5)	0.60 (0.50-0.72)	250	24.6 (21.6– 27.5)	1.1 (0.8–1.4)	0.61 (0.49–0.77)
NIPF	350	20.8 (13.0– 28.5)	2.5 (0.7-4.3)	209	21.5 (17.1– 25.9)	1.2 (0.6–1.8)	0.48 (0.38-0.61)	161	16.7 (13.6– 19.7)	0.8 (0.5–1.0)	0.67 (0.50-0.89)

1999 female population 14 years and older: Overall: 15,775,803.

2009 female population 14 years and older: Overall: 18,982,433.

2017 female population 14 years and older: Overall: 21,520,499.

N = 3,439 (*n* = 354 missing data on stab injury).

^{¥¥}N = 2,387 (n = 20 missing data on stab injury).

* No imputation on overall descriptive summaries.

CI, confidence interval; IPF, intimate partner femicide; IRR, incidence rate ratio; NIPF, non-intimate partner femicide.

https://doi.org/10.1371/journal.pmed.1004330.t004

Characteristic		IPF		NIPF				
	1999 N = 1,610 % (95% CI)	2009 N = 1,294 % (95% CI)	2017 N = 1,089 % (95% CI)	1999 N = 1,686 % (95% CI)	2009 N = 971 % (95% CI)	2017 N = 890 % (95% CI)		
Rape femicides	13.6 (8.7–18.4)	14.6 (11.6–17.7)	4.9 (3.3-6.5)	17.0 (9.4–24.6)	28.7 (23.2-34.2)	12.1 (8.8–15.4)		
Perpetrator convicted	29.4 (20.7-38.2)	29.7 (23.7–35.7)	35.7 (32.6-38.7)	25.9 (19.6-32.3)	18.3 (14.2–22.4)	33.5 (30.2-36.8)		
Perpetrator died by suicide	14.0 (9.1–18.8)	14.5 (11.7–17.3)	12.9 (10.9–14.9)	2.5 (0.1-4.8)	2.7 (1.1-4.3)	1.8 (1.1–2.5)		
History of IPV	31.6 (21.9-41.3)	33.2 (27.7-38.6)	28.6 (25.0-32.2)					

Table 5. Aspects of the murder and the case legal outcome over the 3 studies for IPF and NIPF: weighted and imputed data.

IPF, intimate partner femicide; IPV, intimate partner violence; NIPF, non-intimate partner femicide.

https://doi.org/10.1371/journal.pmed.1004330.t005

Murder, victim, and perpetrator characteristics

Table 5 presents further aspects of the IPF and NIPF murders. There was a decrease in the proportion of rape femicides between 2009 and 2017 for both IPF and NIPF, but these remained much more common among the NIPF compared to IPF cases. In 2017, the proportion of overall cases that were closed with the perpetrators convicted of murder was approximately 1 in 3. The proportion of perpetrators convicted was significantly higher for NIPF cases in 2017 compared to 2009. There was no change in the proportion of IPF cases with convictions across the 3 surveys. The profile of perpetrators of NIPF shows acquaintances/friends known by sight to be the most common perpetrators across the 3 studies with strangers the second most common perpetrators (Table 6). Approximately 1 in 7 of the perpetrators of IPF died by suicide within 6 days of the murder. This proportion is unchanged across the 3 data points and is much higher than the proportion of NIPF perpetrators who died by suicide. A history of IPV provided by the police during the interviews shows similar patterns across the surveys, with a slight decrease in 2017, which was not statistically significant.

Discussion

In 2017, the rate of IPF was 4.9 per 100,000 female population, which is more than 4 times the estimated global rate for intimate partner and family-related homicide, and higher than the rate for any other country [2]. However, South Africa is the only country that has collected femicide data with a dedicated national survey, and it is possible that rates in other countries would be shown to be higher with more accurate estimations. More women were murdered by their intimate partner than by other persons in 2009 and 2017. We have shown a decline in all forms of femicide in South Africa between the 1999 and 2009 and 2009 and 2017 studies. The analysis by age group shows that women aged 14 to 44 are most at risk of IPF; however, this age group also benefitted the most from the decrease in IPF, which contrasts with NIPF where the decrease initially impacted all age groups, but between 2009 and 2017 was only seen among older women (45 years and older). There is some evidence that the decline in IPF may have initially benefitted the youngest women most, with a greater decline seeing in women aged 30 to 44 after 2009 than in other age-groups. The decline in IPF was visible across African and Coloured groups but was most notable for Coloured women who had the highest race subgroup rates in 1999 and 2009. The decline for African women for NIPF was not sustained from 2009 to 2017.

We have also shown that the rate in some provinces of the country is exceptionally high, with more than a 4-fold variation in rates among provinces. Notably, in no province is the rate low by international comparison. The provinces with higher rates, in particular Eastern Cape and Kwa-Zulu Natal, are known to have multiple challenges across social, economic, and

Characteristic		IPF		NIPF			
	1999: N = 1,349 % (95% CI)	2009: <i>N</i> = 1,024 % (95% CI)	2017: N = 768 % (95% CI)	1999: N = 1,335 % (95% CI)	2009: N = 768 % (95% CI)	2017: <i>N</i> = 604 % (95% CI)	
Victim perpetrator relationship							
Husband-current	17.1 (11.3–25.1)	17.1 (13.9–21)	17.1 (15.6–18.7)				
Ex-husband	1.3 (0.6–2.7)	0.7 (0.3-2.1)	1.2 (1.1–1.3)				
Cohabitating boyfriend-current	49.6 (39.7-59.5)	23.8 (17.2–32)	19.4 (15.8–23.5)				
Ex cohabitating boyfriend	0.6 (0.3-1.2)	1.1 (0.5–2.5)	0.6 (0.3–1.3)				
Boyfriend/girlfriend-current*	28.3 (21.9-35.9)	47.8 (41.3–54.5)	53.6 (48.7-58.5)				
Ex-boyfriend	2.8 (1.5-5.2)	7.4 (5.8–9.3)	7.2 (5.9–8.9)				
Rejected man proposing a relationship	0.3 (0.1-0.7)	2.0 (1.1-3.5)	1.0 (0.4–2.2)				
Family**				14.2 (8.2–23.5)	18.7 (14.7–23.5)	17.4 (15.1–19.9)	
Friend/known by sight/acquaintance				51.1 (44.4–57.6)	53.6 (48.3-58.9)	42.1 (37.4-46.9)	
Stranger/unknown				33.9 (26.9–41.6)	19.9 (15.3–25.4)	31.8 (27.3–36.7)	
Other***				0.8 (0.5–1.3)	7.8 (5.8–10.5)	8.8 (6.9–11.0)	
Injury setting							
Major urban	46.9 (30.3-64.2)	32.2 (26.2-38.8)	33.4 (28.0-39.3)	44.8 (27.2–63.9)	33.6 (26.4–41.7)	32.8 (29.5-36.4)	
Rural	53.1 (35.8-69.7)	64.2 (57.5-70.4)	65.9 (59.9–71.4)	55.2 (36.1-72.8)	64.1 (56.0-71.4)	67.2 (63.6-70.5)	
Unknown		3.6 (2.5-5.3)	0.7 (0.3–1.7)		2.3 (1.5-3.4)		
Suspected rape murder							
No	88.8 (84.8-91.9)	84.7 (81.3-87.5)	97.1 (96–97.9)	87.2 (80.1-92.0)	68.5 (62.2–74.2)	87.4 (83.2–90.7)	
Yes	11.2 (8.1–15.2)	10.5 (8.3–13.2)	2.9 (2.1-4)	12.8 (8.0–19.9)	27.4 (22.6–32.9)	12.6 (9.3–16.8)	
Missing info		4.8 (3.3–7)			4.1 (2.3–7.1)		
Perpetrator convicted							
No	64.9 (54.9-73.7)	62.5 (55.9-68.7)	49.4 (45.7–53.2)	67.2 (58.7–74.7)	76.9 (71.4–81.6)	50.6 (46.9-54.4)	
Yes	35.1 (26.3-45.1)	37.5 (31.3-44.1)	50.6 (46.8-54.3)	32.8 (25.3-41.3)	23.1 (18.4–28.6)	49.4 (45.6-53.1)	
Perpetrator died by suicide							
No	83.4 (76.7-88.4)	80.8 (76.5-84.5)	86.9 (84.8-88.8)	96.9 (92.1-98.8)	95.6 (92.4–97.5)	97.8 (96.8–98.5)	
Yes	16.6 (11.6-23.3)	18 (14.6–21.9)	12.9 (11.0–15.0)	3.1 (1.2–7.9)	3.2 (1.8-5.7)	1.8 (1.2–2.7)	
Missing info		1.2 (0.5–2.9)	0.2 (0.1-0.6)		1.2 (0.5–2.7)	0.3 (0.1–1.4)	
History of IPV							
No	68.4 (58.0-77.2)	66.9 (61.2–72.1)	68.8 (65.2–72.2)				
Yes	31.6 (22.8-42)	33.1 (27.9–38.8)	28.1 (24.7-31.8)				
Missing info			3.1 (2.3-4.0)				

Table 6. Victim and perpetrator characteristics over the 3 surveys for IPF and NIPF: weighted and non-imputed data.

*Current boyfriend includes same-sex partner: 1999 IF includes 9 same-sex partners; 2009-no same-sex partner reported; 2017–1 same sex partner.

**Family includes mother, father, brother, sister, stepfather/mother's boyfriend, relatives, and in-laws.

***Other includes female perpetrator romantically involved with current/ex-husband/boyfriend (love triangle), current/ex-client/customer, tenant, community members, employee (gardener, caregiver, and domestic worker), and colleague.

CI, confidence interval; IPF, intimate partner femicide; IPV, intimate partner violence; NIPF, non-intimate partner femicide.

https://doi.org/10.1371/journal.pmed.1004330.t006

political sectors [21]. These results are important for planning responses and prevention intervention at the provincial level, and the 2017 estimates provide a baseline to assess future changes. Provincial data can also be used by advocates and provincial government structures to mobilise resources for provinces that requires them the most. We have also seen a decrease in IPF by gunshot between 2009 and 2017, but no change in gun-related NIPF. There was a decrease in femicide by gunshot between 1999 and 2009, which paralleled an overall decrease in gun-related murders among both men and women that has been widely attributed to effective implementation of the Firearm Control Act [22,23]. However, a general increase in

corruption and lawlessness after 2009 has resulted in a relaxation of gun control with greater circulation of licensed and unlicensed firearms [24]. Our findings suggest that these were particularly used in NIPFs, which generally are perpetrated in the context of other crime, and this coincides with an increase in other firearm-related deaths and injuries [22,23]. This shows the value of strict control of gun ownership and licensing in reducing gun-related murder. The increase in the overall murder by stabbing mirrors an increase that has been seen in male homicides [25], the reason for this is unclear.

The decline in femicide cannot be attributed to improved performance of the police and criminal justice system as there was no evidence of this having occurred, and, notably, there was no change in the proportion of IPF in which there was conviction of the perpetrator across the 3 surveys. The decrease in the number of cases in which a perpetrator was identified and the increase in the number of missing dockets are evidence of a decline in the quality of case investigation and management. These problems with policing and undermining of the criminal investigation capabilities are well documented and reflect poor management and corruption [26].

The decrease in femicide overall mirrors changes that have been noted in deaths from interpersonal violence of men. The latter declined from 25,886 in 1997 to 15,854 in 2012, a 52.4% decline in the age-standardised death rate [27]. The changes in femicide in part reflect overall changes in the country that have resulted in a reduction in the use of violence and murder. These were driven by the end of apartheid (1994) and improved social conditions and access to the economy for the previously marginalised majority [28]. However, our analysis has shown that IPF and NIPF are distinct social phenomena, with different patterns of prevalence across the 3 data points, by age group and by race, and they have differences in underlying drivers. For example, between 1999 and 2009, the IRRs show that NIPF declined faster than IPF, and across all age-groups, whereas the IRRs for IPF only declined for women under 45. It was also seen across femicide types for African and Coloured women, but among African women, IPF declined less than NIPF. Between 2009 and 2017, IPF only declined significantly for women aged 30 to 44 years, whereas the greatest decline in NIPF was seen for women aged 45 years and over. Studies have consistently shown younger women to be at higher risk for IPV [29]. The extent of poverty and social norms around the use of violence may impact both IPF and NIPF, but IPF is also importantly driven by IPV and social norms around gender equality [21]. The patterns of change in IPF fit well with what is known about changes in social norms around gender equity and the acceptability of IPV that have occurred in the country, that have seen great reductions in acceptability of IPV from 1998 to 2016, but more traditional views on the acceptability of wife beating held by African women living in rural settings may contribute to slower change [15].

The femicide surveys are currently the only national data on IPV for South Africa. They provide the most comprehensive insights into the likely impact of national efforts to eliminate violence against women that have been conducted over the last 2 decades. The research has been conducted against a backdrop of a decrease in the overall rate of murder (South African legal term) in the country since 1994 [14]. While this led to a reduction in female murder overall, it raised a key question about whether the murder of women by intimate partner perpetrators would be just as preventable as murder by non-intimate partners. In South Africa, combatting violence against women has been a national priority since the advent of democracy in 1994. One of the first laws passed by the Government was the Domestic Violence Act, signed into law in 1998, which included provisions for access to protection orders [30]. The law was a major focus of activism in 1998 led by the Soul City Institute and linked to an edutainment television series on domestic violence, which at the time had 8 million viewers. In the early 2000s, there was considerable activism around demands for public access to postexposure

prophylaxis for HIV after rape. Activism around rape heightened from 2005 to 2006 when the then Deputy-President, Jacob Zuma, was charged and tried for rape. As he defended himself, rape and violence against women became a major national focus for debates and anti-rape activism. This provided impetus for the 2007 amendments to the Sexual Offences legislation, which substantially widened the definition of rape [30]. In 2013 and 2014, there was a similar national focus on femicide when the para-Olympic sprinter Oscar Pistorius was charged with, and tried for, the murder of his girlfriend. Gender-based violence activism continued and came to the fore in 2018, resulting in the President convening a national summit and establishing a process that led to the adoption of a National Strategic Plan to Combat Gender-based Violence and Femicide (GBVF) in 2020 [13]. Throughout these years, violence against women has been kept in the public eye by the media and women's organisations especially during Women's Month in August and the 16 days of Activism in November each year, as well as the periodic release of research results, and many Government Departments have developed strategies for addressing it. The South African experience of feminist activism that led to genderbased violence (GBV) policy change is not unique. Indeed, in a systematic review by Weldon and Htun [31], the authors reviewed the GBV policies of 70 countries and found women's right movements were the most consistent factor that brought about GBV policy change.

Thus, it seems likely that the decrease in IPF indicates progress in reducing IPV in South Africa. Despite the considerable amount of research on IPV conducted over more than 30 years, which include prevalences studies among women as victims and men as perpetrators [32,33], studies within marginalised and high risk groups [34,35], risk factor studies [36], the large body of research on development and testing of primary prevention interventions [37,38], and the impact of IPV on women's health (e.g., HIV, mental health) [39-41], the country has no surveillence through national IPV prevalence surveys, and the research on IPF represents the only IPV surveillance within the country. We do not know if the decline in IPF mirrors a similar decline in all forms of IPV in the country over the 18 years, but we would expect that there should have been a decrease in IPV for this to have been observed. Given the broad range of responses to GBV, it is impossible to pinpoint moments when activities contributed to the decline in femicide; rather, it is likely that the sum total has influenced a decline in violence, as has been shown in other countries [42]. This is in line with a review of GBV policy change in 70 countries that showed the crtical role of local feminist movements in changing patriarchal gender norms [31]. Many high-income countries have shown a decrease in femicide at a population level [8]; however, to the best of our knowledge, this is the first time that similar declines have been described through national surveillance in a middle-income country. An important conclusion of our research, taken together with that from other countries, is that IPF is preventable at a population level, and we suspect it provides further evidence that real shifts in population prevalence of IPV can occur when countries focus on it as a national problem.

Our study's strength is our use of a consistent methodology conducted over 3 studies spanning 18 years and which was initially based on a study design developed by author Martin [43]. We also show the value of dedicated surveys in the absence of functional information systems, and we support the call from UN bodies for focused attention to the collection of data to improve government responses to killing of women and to allow for comparisons across communities and regions [44]. A study limitation includes the variability in sampling across the 3 years; however, it is very unclear what the impact of this would have been. We also considered the role of missing data in our estimates. Firstly, it is possible that we missed murder cases that were not processed by the official MLMs of the country or we missed cases that were misclassified as undetermined cause of death. We, however, believe this would have been only a few cases as all unnatural deaths in the country must undergo a postmortem examination and a recent injury mortality report show reasonable congruency between murders reported by SAPS and forensic data [25]. Undetermined cause of death is a well-described phenomenon of death certification in South Africa, and the 2017 Injury Mortality Study reported 11.2% of unnatural deaths remaining undetermined. A second aspect of missing data are the cases not traced in the police system. The proportion of cases in 2017 that were not in the police system was higher than the proportion in 2009, although similar to that in 1999. Access to the police was more challenging in the 2017 study due to COVID-19, as this limited the research team's ability to visit the police stations and assist the police staff in locating dockets, and this may have contributed to higher number of non-traceable police cases. Although we relied on the CAS number, we discovered many challenges with CAS numbers being changed or documented incorrected (inconsistent allocation of CAS numbers), and we requested ethical approval to use ID numbers as an additional tracing tool. For some cases where a SAPS investigation was not identified, i.e., a CAS and ID number not found on the police data systems, we requested SAPS to open an investigation. This, however, happened at the end of the study, and any further perpetrator data identified by the police could not be included in our database. The inability to not trace the investigation of cases within the police system (15.7% in 2017) may have contributed to the underestimation of IPF and NIPF as these cases were not included in this analysis. The third aspect of missing data related to cases where a police investigation had occurred, but the investigation process was unable to identify a perpetrator. This may also have had a potential impact on the estimates of NPF and NIPF. We used imputation (details described in methods) as this approach was robust since we had very good information on a large number of cases. We calculated rates on non-imputed data and a similar pattern of decline across the 3 surveys was found for IPF and NIPF. (See S2 and S3 Tables for non-imputed results.) The improvement of an integrated information system with linkages between data sets (police, justice, forensic, etc.) is a key strategic objective in the recently developed Femicide Prevention Strategy developed for South Africa. This strategy is awaiting final Government approval [3].

We also mostly used the population data from the Thembisa model for calculation of the ASR, but because of unavailability of race population data, we used the Statistics South Africa midyear population data for the calculation of rates by race group. There were minimal differences between overall population data across the 2 population data sources (Table 2 footnote).

We also acknowledge that the suspect identified by the police may not have always been the one responsible in some cases, but we have been consistently aware of this possibility in our research and have formed our opinion on whether a women was killed by a partner or a non-partner based on a description of circumstances of the death. It is well known that convictions are not possible in cases with insufficient evidence and depending on a final court outcome will result in underestimations [26]. Our estimates for 1999 and 2009 differ very slightly from what has been previously published without using imputation. However, the non-imputed data presented in the supplemental file shows similar patterns and statistically significant reductions between studies for most of the analysis. We believe imputation allowed us to get closer to describing the full dimension of femicide in the country.

We have shown that South Africa has the highest globally recorded rate of femicide, and yet there have been large declines in all types of femicide over a period spanning 18 years. This shows that femicide can very likely be impacted by important social dynamics, including overall increases in political stability and less general violence, as well as economic well-being, but furthermore that activism, policies, and programmes to build greater gender equity and combat the violence against women, can result in important reductions in IPF, and we suspect underlying levels of IPV. The public health implications point to the huge importance of population-level structural interventions in reducing levels of femicide as well as the need to support feminist movements and community activism against violence against women.

Supporting information

S1 STROBE Checklist. STROBE statement. (DOCX)

S1 Text. Research protocol. (PDF)

S1 Table. Count and percentages distribution of all female murder cases in the police system by mortuary strata across the 3 surveys. (DOCX)

S2 Table. Age-standardised population rates for 1999, 2009, and 2017 for overall female murders: IPF and NIPF by age and race and IRRs of population rate estimates between surveys: weighted and non-imputed data. IPF, intimate partner femicide; IRR, incidence rate ratio; NIPF, non-intimate partner femicide. (DOCX)

S3 Table. Age-standardised population rates for 1999, 2009, and 2017 for all female murders: IPF and NIPF by death mechanism and incidence. IPF, intimate partner femicide; NIPF, non-intimate partner femicide. (DOCX)

Acknowledgments

We thank the National and Provincial Departments of Health and the various Forensic Pathology Services for providing access to the mortuaries. We also thank the National and Provincial Police Services for their support as well as the many Investigating Officers who assisted us in the collecting of the data. Finally, we thank the team of fieldworkers for their tireless work in data collection.

Author Contributions

Conceptualization: Naeemah Abrahams, Lorna J. Martin, Rachel Jewkes.

- **Data curation:** Naeemah Abrahams, Shibe Mhlongo, Esnat Chirwa, Bianca Dekel, Asiphe Ketelo, Carl Lombard, Megan Prinsloo.
- **Formal analysis:** Naeemah Abrahams, Shibe Mhlongo, Esnat Chirwa, Bianca Dekel, Asiphe Ketelo, Carl Lombard, Leane Ramsoomar, Shanaaz Mathews, Gérard Labuschagne, Richard Matzopoulos, Megan Prinsloo, Rachel Jewkes.

Funding acquisition: Naeemah Abrahams, Rachel Jewkes.

Investigation: Naeemah Abrahams.

- Methodology: Naeemah Abrahams, Shibe Mhlongo, Esnat Chirwa, Carl Lombard, Lorna J. Martin, Rachel Jewkes.
- **Project administration:** Naeemah Abrahams, Bianca Dekel, Asiphe Ketelo, Nwabisa Shai, Leane Ramsoomar.
- Supervision: Bianca Dekel, Asiphe Ketelo, Carl Lombard, Nwabisa Shai, Leane Ramsoomar, Gérard Labuschagne, Lorna J. Martin, Rachel Jewkes.
- Validation: Shibe Mhlongo, Esnat Chirwa, Bianca Dekel, Asiphe Ketelo, Carl Lombard, Shanaaz Mathews, Megan Prinsloo, Lorna J. Martin, Rachel Jewkes.

Visualization: Shibe Mhlongo, Esnat Chirwa.

Writing – original draft: Naeemah Abrahams.

Writing – review & editing: Naeemah Abrahams, Shibe Mhlongo, Esnat Chirwa, Bianca Dekel, Asiphe Ketelo, Carl Lombard, Nwabisa Shai, Leane Ramsoomar, Shanaaz Mathews, Gérard Labuschagne, Richard Matzopoulos, Megan Prinsloo, Lorna J. Martin, Rachel Jewkes.

References

- Stöckl H, Devries K, Rotstein A, Abrahams N, Campbell J, Watts C, et al. The global prevalence of intimate partner homicide: a systematic review. Lancet. 2013; 382(9895):859–865. <u>https://doi.org/10. 1016/S0140-6736(13)61030-2 PMID: 23791474</u>
- 2. United Nations Office on Drugs and Crime. Global Study On Homicide: Gender-related killing of women and girls. Vienna: UNODC Research; 2019. Available from: https://www.unodc.org/documents/data-and-analysis/gsh/Booklet_5.pdf (Last retrieved 15 Jun 2023).
- Shai N, Ramsoomar L, Abrahams N. Femicide Prevention Strategy Development Process: The South African Experience. Peace Rev. 2022; 34(2):227–245. https://doi.org/10.1080/10402659.2022. 2049001
- Zecha A, Abrahams N, Duhamel K, Fabre C, Otamendi A, Cazares AR, et al. Data sources and challenges in addressing femicide and feminicide. In: Dawson M, Vega SM, editors. The Routledge International Handbook on Femicide and Feminicide. Routledge; 2023.
- Abrahams N, Mathews S, Martin LJ, Lombard C, Jewkes R. Intimate partner femicide in South Africa in 1999 and 2009. PLoS Med. 2013; 10(4):e1001412. https://doi.org/10.1371/journal.pmed.1001412 PMID: 23565064
- Zahn MA. Intimate partner homicide: An overview. NIJ J. 2003;(250):2–3. Available from: https://www. ojp.gov/pdffiles1/jr000250.pdf. (Last retrieved 15 Jun 2023).
- National Homicide Monitoring Program. Directory of Family, Domestic, and Sexual Violence Statistics, 2018. Available from: https://www.aihw.gov.au/reports/domestic-violence/family-domestic-and-sexualviolence (Last retrieved 15 Jun 2023).
- 8. United Nations Office on Drugs and Crime. Gender related killings of women and girls (femicide/feminicide): Global estimates of gender-related killings of women and girls in the private sphere in 2021 UN Women 2022. Available from: https://www.unwomen.org/en/digital-library/publications/2022/11/gender-related-killings-of-women-and-girls-improving-data-to-improve-responses-to-femicide-feminicide (Last retrieved 15 Jun 2023).
- Inquests Act. Act No 58 of 1959, as amended. 1959 (1959). Available from: https://www.gov.za/sites/ default/files/gcis_document/201505/act-58-1959.pdf (Last retrieved 14 Nov 2023).
- Vetten L. "Man shoots wife": Intimate femicide in Gauteng, South Africa 1996" Crime and Conflict 6 (Winter): 1–6. Available from: https://help.csvr.org.za/man-shoots-wifeq-intimate-femicide-in-gautengsouth-africa/ (Last retrieved 4 Oct 2023).
- Harris PA, Taylor R, Thielke R, Payne J, Gonzalez N, Conde JG. Research electronic data capture (REDCap)–A metadata-driven methodology and workflow process for providing translational research informatics support. J Biomed Inform. 2009; 42(2):377–81. https://doi.org/10.1016/j.jbi.2008.08.010 PMID: 18929686
- Bureau of Justice Statistics. Uniform Crime Report: Crime in the United states, 2019 Murder US FBI; 2019. Available from: https://ucr.fbi.gov/crime-in-the-u.s/2019/crime-in-the-u.s.-2019 (Last retrieved 15 Jun 2023).
- Republic of South Africa. National Strategic Plan on Gender-based Violence & Femicide Pretoria South African Government; 2020. Available from: www.justice.gov.za/vg/gbv/nsp-gbvf-final-doc-04-05.pdf (Last rerieved 15 Jun 2023).
- South African Police Service. Annual Crime Statistics Pretoria: South African Police Service SAP; 2020. Available from: https://www.saps.gov.za/services/crimestats.php (Last retrieved 4 Oct 2023).
- National Department of Health (NdoH), Statistics South Africa (Stats SA), South African Medical Research Council. South African Demographic and Health Survey 2016. Pretoria: National Department of Health; 2019. Available from: https://dhsprogram.com/pubs/pdf/FR337/FR337.pdf (Last retrieved 3 Oct 2023).

- Dorrington R. Alternative South African mid-year estimates. Cape Town Centre for Actuarial Research; 2013. Available from: http://www.care.uct.ac.za/sites/default/files/image_tool/images/561/Downloads/ Mono11.pdf (Last retrieved 15 Jun 2023).
- Johnson LF, May MT, Dorrington RE, Cornell M, Boulle A, Egger M, et al. Estimating the impact of antiretroviral treatment on adult mortality trends in South Africa: A mathematical modelling study. PLoS Med. 2017; 14(12):e1002468. https://doi.org/10.1371/journal.pmed.1002468 PMID: 29232366
- Ahmad OB, Boschi-Pinto C, Lopez AD, Murray CJ, Lozano R, Inoue M. Age standardization of rates: a new WHO standard. Geneva: World Health Organization. 2001; 9(10):1–14.
- Statistics South Africa. Mid-year population estimates, 2022. Pretoria: Statistics South Africa; 2022. Available from: https://www.statssa.gov.za/publications/P0302/P03022022.pdf (Last retrieved 15 Jun 2023).
- Andridge RR, Little RJA. A Review of Hot deck Imputation for Survey Non-response. Int Stat Rev. 2010; 78(1):40–64. https://doi.org/10.1111/j.1751-5823.2010.00103.x PMID: 21743766
- Statistics South Africa. General Household Survey 2021. Available from: https://www.statssa.gov.za/? p=15482 (Last retrieved 15 Jun 2023).
- Matzopoulos R, Simonetti J, Prinsloo M, Neethling I, Groenewald P, Dempers J, et al. A retrospective time trend study of firearm and non-firearm homicide in Cape Town from 1994 to 2013. SAMJ.2018; 108(3):197–204. https://doi.org/10.7196/SAMJ.2018.v108i3.12756 PMID: 30004363
- 23. Matzopoulos R, Prinsloo M, Bradshaw D, Abrahams N. Reducing homicide through policy interventions: The case of gun control. SAMJ. 2019; 109(11b):63. <u>https://doi.org/10.7196/SAMJ.2019.v109i11b.</u> 14256 PMID: 32252871
- 24. Rapanyane MB. Seizure of State Organs, Corruption and Unaccountability Promotion in South Africa: Case Study of Jacob Zuma Administration. Afr J Dev Stud. 2021; 11(3):251–70. https://doi.org/10. 31920/2634-3649/2021/v11n3a12
- 25. Prinsloo M, Mhlongo S, Dekel B, Gwebushe N, Martin LJ, Saayman G, et al. The 2nd Injury Mortality Survey: A national study of injury mortality levels and causes in South Africa in 2017. Cape Town South African Medical Research Council. 2021. Available from: https://www.samrc.ac.za/research-reports/ 2nd-injury-mortality-survey-national-study-injury-mortality-levels-and-causes?type=researchreport&menu=intramural-research-units&menu-item=burden of disease (Last retrieved 15 Nov 2023).
- 26. Labuschagne G. The profiler's Diaries- From the Case Files of a Police Psychologist. Cape Town: Penguin Random House; 2021.
- Pillay-van Wyk V, Msemburi W, Laubscher R, Dorrington RE, Groenewald P, Glass T, et al. Mortality trends and differentials in South Africa from 1997 to 2012: second National Burden of Disease Study. Lancet Glob Health. 2016; 4(9):e642–e653. <u>https://doi.org/10.1016/S2214-109X(16)30113-9</u> PMID: 27539806
- Coovadia H, Jewkes R, Barron P, Sanders D, McIntyre D. The health and health system of South Africa: historical roots of current public health challenges. Lancet. 2009; 374(9692):817–834. <u>https://doi.org/10.1016/S0140-6736(09)60951-X PMID: 19709728</u>
- Sardinha L, Maheu-Giroux M, Stöckl H, Meyer SR, García-Moreno C. Global, regional, and national prevalence estimates of physical or sexual, or both, intimate partner violence against women in 2018. Lancet. 2022; 399(10327):803–813.33. <u>https://doi.org/10.1016/S0140-6736(21)02664-7</u> PMID: 35182472
- Mogale RS, Burns KK, Richter S. Violence against women in South Africa: Policy position and recommendations. Violence Against Women. 2012; 18(5):580–594. <u>https://doi.org/10.1177/1077801212453430 PMID: 22826009</u>
- **31.** Weldon SL, Htun M. Feminist mobilisation and progressive policy change: why governments take action to combat violence against women. Gend Dev. 2013; 21(2):231–247.
- Jewkes R, Penn-Kekana L, Levin J, Ratsaka M, Schrieber M. Prevalence of emotional, physical and sexual abuse of women in three South African provinces. SAMJ. 2001; 91(5):421–428. PMID: 11455808
- Abrahams N, Jewkes R, Laubscher R, Hoffman M. Intimate partner violence: Prevalence and risk factors for men in Cape Town, South Africa. Violence Vict. 2006; 21(2):247–263. https://doi.org/10.1891/ vivi.21.2.247 PMID: 16642742
- Coetzee J, Gray GE, Jewkes R. Prevalence and patterns of victimization and polyvictimization among female sex workers in Soweto, a South African township: a cross-sectional, respondent-driven sampling study. Glob Health Action. 2017; 10(1):1403815. <u>https://doi.org/10.1080/16549716.2017.1403815</u> PMID: 29211633
- **35.** Gibbs A, Dunkle K, Jewkes R. Emotional and economic intimate partner violence as key drivers of depression and suicidal ideation: A cross-sectional study among young women in informal settlements

in South Africa. PLoS ONE. 2018; 13(4):e0194885. https://doi.org/10.1371/journal.pone.0194885 PMID: 29659595

- Jewkes R, Levin J, Penn-Kekana L. Risk factors for domestic violence: findings from a South African cross-sectional study. Soc Sci Med. 2002; 55(9):1603–1617. https://doi.org/10.1016/s0277-9536(01) 00294-5 PMID: 12297246
- Jewkes R, Nduna M, Levin J, Jama N, Dunkle K, Puren A, Duvvury N. Impact of Stepping Stones on incidence of HIV, HSV-2 and sexual behaviour in rural South Africa: cluster randomised controlled trial. BMJ. 2008; 337:a506.
- Gibbs A, Abdelatif N, Washington L, Chirwa E, Willan S, Shai N, et al. Differential impact on men in an IPV prevention intervention: A post hoc analysis using latent class analysis of the Stepping Stones and Creating Futures intervention in South Africa. Soc Sci Med. 2020; 265:113538. https://doi.org/10.1016/ j.socscimed.2020.113538 PMID: 33257178
- Mhlongo S, Seedat S, Jewkes R, Myers B, Chirwa E, Nöthling J, et al. Depression and post-traumatic stress symptoms two years post-rape and the role of early counselling: Rape Impact Cohort Evaluation (RICE) study. Eur J Psychotraumatol. 2023; 14(2):2237364. https://doi.org/10.1080/20008066.2023. 2237364 PMID: 37642373
- Machisa MT, Christofides N, Jewkes R. Mental ill health in structural pathways to women's experiences of intimate partner violence. PLoS ONE. 2017; 12(4):e0175240. https://doi.org/10.1371/journal.pone. 0175240 PMID: 28384241
- 41. Jewkes R, Dunkle K, Nduna M, Shai N. Intimate partner violence, relationship gender power inequity, and incidence of HIV infection in young women in South Africa: a cohort study. Lancet. 2010; 367:41–8.
- **42.** Ellsberg M, Quintanilla M, Ugarte WJ. Pathways to change: Three decades of feminist research and activism to end violence against women in Nicaragua. Glob Public Health. 2022; 17(11):3142–3159. https://doi.org/10.1080/17441692.2022.2038652 PMID: 35184690
- 43. Martin L. Violence against women: an analysis of the epidemiology and patterns of injury in rape homicide in Cape Town and in rape in Johannesburg [unpublished MMed forensic pathology thesis]. University of Cape Town; 1999. Available from: <u>https://open.uct.ac.za/handle/11427/30501</u> (last retrieved 23 Jun 2023).
- 44. United Nations Office on Drugs and Crime. Statistical framework for measuring the gender-related killing of women and girls (also referred to as "femicide/feminicide"). United Nations. Available from: https://www.unodc.org/documents/data-andanalysis/statistics/Statistical_framework_femicide_2022. pdf (Last retrieved 2 Oct 2023).